Author's response to reviews

Title: Pitfalls in the statistical examination and interpretation of the correspondence between physician and patient satisfaction ratings and their relevance for shared decision making research

Authors:

Oliver Hirsch (oliver.hirsch@staff.uni-marburg.de)
Heidemarie Keller (kellerhe@staff.uni-marburg.de)
Christina Albohn-Kühne (christina.albohn@staff.uni-marburg.de)
Tanja Krones (tanja.krones@usz.ch)
Norbert Donner-Banzhoff (norbert@staff.uni-marburg.de)

Version: 6 Date: 13 May 2011

Author's response to reviews:

Table 3. Comparison of physician and patient satisfaction ratings in the intervention group with the Wilcoxon signed-rank test. We applied Bonferroni correction for multiple testing (#= .05/6 = .008). Item means and standard deviations and means, standard deviations, and medians of the differences are listed. The scale ranges from 1 (totally agree) to 5 (totally disagree).